Author's response to reviews

Title: Quantification of Bone Marrow Lesion Volume and Volume Change Using Semi-automated Segmentation: Data from the Osteoarthritis Initiative

Authors:

Jincheng Pang (pangjc@eecs.tufts.edu)
Jeffrey B Driban (jdriban@tuftsmedicalcenter.org)
Geoffroy Destenaves (GDestenaves@tuftsmedicalcenter.org)
Eric Miller (elmiller@ece.tufts.edu)
Grace H Lo (ghlo@bcm.edu)
Robert J Ward (rward@tuftsmedicalcenter.org)
Lori Lyn Price (lprice1@tuftsmedicalcenter.org)
John A Lynch (jlynch@psg.ucsf.edu)
Charles B Eaton (charles_eaton@mhri.org)
Timothy E McAlindon (tmcalindon@tuftsmedicalcenter.org)

Version: 2 Date: 20 July 2012

Author's response to reviews:

Please see the original cover letter dated July 16, 2012.